Author’s response to reviews

Title: What is in the drug packet? Access and use of non-prescribed poly-pharmaceutical packs (Yaa Chud) in the community in Thailand

Authors:

Malee Sunpuwan (malee.sun@mahidol.edu)
Sureeporn Punpuing (sureeporn.pun@mahidol.ac.th; prsureeporn@gmail.com)
Wipaporn Jaruruengpaisan (wipaporn.jar@mahidol.ac.th)
John Kinsman (john.kinsman@epiph.umu.se)
Heiman Wertheim (Heiman.Wertheim@radboudmc.nl)

Version: 2 Date: 29 Jun 2019

Author’s response to reviews:

Response to an assistant Editor Comments:

1. Image Attribution
   - Removed additional files 1 and 2.
   - In the text:
     = Line 77-78, we added (for more details, see http://pca.fda.moph.go.th/public_media_detail.php?id=6&cat=40&content_id=835)
     = Line 81, we added (for more details, see https://www.hfocus.org/content/2017/04/13780)

2. Consent for Publication
   - Changed to “Not Applicable” as suggested.

3. Abbreviations
4. Clean Manuscript

- Based on your advice, we have checked details before we uploaded a clean manuscript as well as figures and additional files.